Author's response to reviews

Title: Why do adult women in Vietnam take iron tablets?

Authors:
Ritsuko Aikawa (ritsuko_aikawa@hotmail.com)
Masamine Jimba (mjimba@m.u-tokyo.ac.jp)
Khan C Nguyen (nckhan@hn.vnn.vn)
Yun Zhao (Y.Zhao@exchange.curtin.edu.au)
Colin W Binns (C.Binns@curtin.edu.au)
Mi Kyung Lee (M.K.Lee@murdoch.edu.au)

Version: 2  Date: 18 November 2005

Author's response to reviews:

Dear Editor

The following changes are made:

Page original revised
1 Rituko Ritsuko
4 Introduction Background
15-16 Competing interests

The findings and conclusion in this article are those of authors and do not necessarily reflect the view of the funding agency, Japan International Cooperation Agency, nor the organizations where authors currently work (Tokyo University, National Institute of Nutrition, Curtin University of Technology, and Murdoch University, respectively). There were no competing interests financially or otherwise, in the present investigation, and no incentive on behalf of the Japan International Cooperation Agency to obtain the results found. No fees or funding were obtained from any organizations that could gain or lose from the publication of the manuscript, and no stock or shares are held in any organizations that stand to gain or lose financially from this publication.

16 The contents of this study are the authors and do not necessarily reflect the view of JICA and National Nutrition Institute of Vietnam.

Thank you for your attention.

Mi Kyung Lee